Author's response to reviews

Title: Efficacy and safety of acupuncture for the treatment of non-specific acute low back pain: a randomised controlled multicentre trial protocol [ISRCTN65814467]

Authors:

Jorge Vas (jvas@acmas.com)
Emilio Perea-Milla (eperea@hcs.es)
Camila Mendez (camila.mendez@juntadeandalucia.es)
Luis Carlos Silva (lcsilva@infomed.sld.cu)
Antonia Herrera Galante (ahgalante@gmail.com)
Jose Manuel Aranda Regules (regules@samfyc.es)
Dulce M Martinez Barquin (dulma@inicia.es)
Inmaculada Aguilar (iaguilar42@mixmail.com)
Vicente Faus (vfaus@hcs.es)

Version: 5 Date: 17 April 2006

Author's response to reviews: see over
Dear Sir/Madam,

Point-by-point description of the changes made

Abstract – Discussion section is added

Manuscript sections - Manuscript sections include (in the following order): Abstract; Background; Methods/Design; Discussion; Abbreviations; Competing interests; Authors’ contributions; Acknowledgements; References; and Figure legends.

Yours faithfully,

Dr Jorge Vas